

WORKING PAPER 218

Optimal fiscal policy and sovereign debt crises

Stefan Niemann and Paul Pichler

The Working Paper series of the Oesterreichische Nationalbank is designed to disseminate and to provide a platform for discussion of either work of the staff of the OeNB economists or outside contributors on topics which are of special interest to the OeNB. To ensure the high quality of their content, the contributions are subjected to an international refereeing process. The opinions are strictly those of the authors and do in no way commit the OeNB.

The Working Papers are also available on our website (http://www.oenb.at) and they are indexed in RePEc (http://repec.org/).

Publisher and editor	Oesterreichische Nationalbank
	Otto-Wagner-Platz 3, 1090 Vienna, Austria
	PO Box 61, 1011 Vienna, Austria
	www.oenb.at
	oenb.info@oenb.at
	Phone (+43-1) 40420-6666
	Fax (+43-1) 40420-046698
Editorial Board of the Working Papers	Doris Ritzberger-Grünwald, Ernest Gnan, Martin Summer
Coordinating editor	Martin Summer
Design	Communications and Publications Division
DVR 0031577	
ISSN 2310-5321 (Print) ISSN 2310-533X (Online)	

© Oesterreichische Nationalbank, 2017. All rights reserved.

Optimal fiscal policy and sovereign debt crises^{*}

Stefan Niemann^{\dagger} Paul Pichler^{\ddagger}

December 2017

Abstract

This paper studies how sovereign risk – both fundamental and self-fulfilling – shapes the cyclical behavior of optimal fiscal policy. We develop a model with endogenous default costs where market sentiment can induce belief-driven debt rollover crises. Optimal taxes and public spending are generally procyclical, but the incidence of rollover risk gives rise to infrequent episodes of severely countercyclical fiscal activity. These endogenous regime changes are associated with pronounced countercyclical changes in the level of debt. Debt buildups are triggered already by relatively mild recessions, but successful fiscal consolidations occur only in exceptionally good times.

JEL classification: E62, F34, H63 Keywords: fiscal policy; sovereign debt; rollover crises; regime switches

^{*}Earlier versions of this paper were presented at the RES 2016 Annual Conference (Sussex) and CEF2016 (Bordeaux). We would like to thank seminar audiences as well as James Costain, Morten Ravn and Almuth Scholl for helpful discussions. The views expressed in this paper are those of the authors and do not necessarily reflect the views of the Oesterreichische Nationalbank or the European System of Central Banks.

[†]Department of Economics, University of Essex, United Kingdom. E-mail: sniem@essex.ac.uk

[‡]Economic Studies Division, Oesterreichische Nationalbank, and Department of Economics, University of Vienna, Austria. E-mail: paul.pichler@oenb.at

Non-technical Summary

We study how the presence of sovereign risk - both fundamental solvency risk and sentimentdriven liquidity risk - affects the cyclical behavior of optimal fiscal policy and the pattern of government debt issuance. We develop a dynamic economic model with distortionary taxation, government spending and endogeneous costs of government default arising from disruptions to the import of intermediate inputs. Market sentiment can give rise to self-fullfilling debt rollover crises when international lenders lose confidence in the government's willingness to honor its liabilities.

Optimal taxes and public spending are generally *procyclical* in our environment, which is well in line with empirical evidence from emerging economies where fiscal policy often does not contribute to smoothing business cycle fluctuations over time but rather exacerbates underlying boom-bust cycles. However, the incidence of rollover risk gives rise to occasional episodes of severely countercyclical fiscal activity. These episodes occur when the economy enters or leaves the critical region of indebtedness where self-fulfilling rollover crises are possible.

The government's optimal fiscal policy during recessions can then be expansionary. In order to avoid an excessive depression in output and consumption, the government optimally exploits its fiscal space even though this policy ultimately entails a discontinuous increase in sovereign risk and future borrowing costs. By contrast, if the level of government debt is initially very high, a sustained economic boom can induce the government to adopt a severely contractionary fiscal stance. It optimally reduces public spending and raises taxes in an effort to bring down debt to the point where rollover risk is eliminated. This policy lowers the default premia charged by international creditors and hence significantly reduces borrowing costs, which provides scope for future governments to reduce distortionary taxes.

In addition to prescribing large and fast-paced debt dynamics, our results also uncover a fundamental asymmetry underlying the optimal fiscal policy: Large fiscal expansions are triggered already by relatively mild but sustained recessions, whereas large fiscal contractions are triggered only by exceptional boom episodes.

1 Introduction

A well-established stylized fact in international macroeconomics is the procyclical pattern of fiscal policy in emerging economies. Different from advanced economies, fiscal policy in these countries does not contribute to smoothing business cycle fluctuations over time but rather exacerbates underlying boom-bust cycles. A common explanation for this phenomenon is that governments in emerging countries have limited access to international borrowing, and particularly so in bad times. Hence, in order to confront concerns about the sustainability of their public finances and to retain financial market access, they often impose contractionary fiscal measures even during severe recessions (cf. Vegh and Vuletin, 2015; Born, Mueller, and Pfeifer, 2015). This fiscal procyclicality can be rationalized insofar as austerity in the form of spending cuts or tax hikes helps to create better borrowing conditions as reflected by reduced sovereign spreads. Importantly, however, these spreads are not only determined by fundamentals but also subject to market sentiment (Calvo, 1988; Cole and Kehoe, 2000).¹

In this paper we study how fundamental sovereign risk and market sentiment shape the pattern of optimal debt issuance and the cyclical behavior of optimal fiscal policy. We develop a model with distortionary taxation, government spending and endogeneous default costs driven by disruptions to the import of intermediate inputs (cf. Mendoza and Yue, 2012). Market sentiment can give rise to belief-driven debt rollover crises when international lenders lose confidence in the government's willingness to honor its liabilities and hence refuse rolling over their debts (cf. Cole and Kehoe, 2000). When debt is sufficiently high and repayment of the maturing liabilities is hence sufficiently costly, the inability to issue new debt gives the government an incentive to default, thus making the investors' initial loss of confidence a self-fulfilling prophecy. Debt levels for which these self-fulfilling dynamics are possible fall into the so-called *crisis zone.*²

¹For example, a narrative advanced in the context of the recent European debt crisis has been that a (sudden) loss of confidence in some countries' capacity to serve their debt triggered a collapse of bond prices; this led to the adoption of severe austerity measures, which further contracted economic activity. Similarly, the Mexican debt crisis of 1994/95 saw the Mexican government unable to roll over its debt because of international investors' pessimistic beliefs and was resolved only through an international rescue package (cf. Cole and Kehoe, 1996).

 $^{^{2}}$ In addressing self-fulfilling expectations as a source of macroeconomic instability, our multiple-equilibria approach is related to, but conceptually distinct from, work on local (in)determinacy. For important applications dealing with the interaction of sovereign risk and fiscal policy, see e.g. Corsetti, Kuester, Meier, and Mueller (2013, 2014).

At the heart of the fiscal policy problem is a trade-off between the government's motives to frontload and to smooth consumption: Increasing current consumption comes at the cost of accumulating more debt and thus higher exposure to default risk, which makes future consumption more volatile. To what extent this happens actually depends on the sources of sovereign risk, and this is reflected in the pattern of optimal fiscal policy. We show that, while optimal taxes and public spending are generally procyclical,³ the incidence of rollover risk gives rise to occasional episodes of severely countercyclical fiscal activity. These episodes are manifestations of *endogenous regime switches* when the economy enters or leaves the crisis zone. Below the crisis zone, the government's optimal fiscal policy during recessions is then expansionary. It exploits its fiscal space in order to avoid an excessive depression in output and consumption, even though this policy ultimately entails a transition into the crisis zone and hence a discontinuous increase in sovereign risk and future borrowing costs. By contrast, within the crisis zone, a sustained boom can induce the government to adopt a severely contractionary fiscal stance. It reduces public spending and raises taxes in an effort to bring down debt to the point where rollover risk is completely eliminated. This policy lowers the default premia charged by international creditors and hence significantly reduces borrowing costs, which provides space for future governments to reduce distortionary taxes. Fiscal austerity, if initiated during booms, can thus be expansionary and facilitate consumption smoothing. The traditional rationale for countercyclical fiscal policy familiar from advanced economies is therefore at work also in our framework where fiscal policy is constrained by both solvency considerations and market sentiment.

In addition to prescribing large and fast-paced debt dynamics during regime switches, our results also uncover a fundamental asymmetry underlying the optimal fiscal policy: Large fiscal expansions are triggered already by relatively mild (but sustained) recessions, whereas large fiscal contractions are triggered only by exceptional boom episodes, that is, when output is significantly above trend. Our normative results thus also facilitate a new perspective at empirically observed debt dynamics. Absent rollover risk, Eaton-Gersovitz-type models prescribe a procyclical debt policy, so that debt rises (falls) in good (bad) times. The empirical evidence, however, indicates that *large swings* in public debt display a different, countercyclical pattern

 $^{^{3}}$ Throughout, we refer to procyclical fiscal policy as implying higher (lower) public spending and lower (higher) tax rates in good (bad) times.

(Abbas, Belhocine, El-Ganainy, and Horton, 2011).⁴ The endogenous regime switches in our model are consistent with this and also with the finding that debt-consolidating fiscal austerity programs pay off particularly if initial economic conditions are benign (Born, Mueller, and Pfeifer, 2015).

Our paper also connects to an expanding literature investigating the role and properties of fiscal policy in environments subject to sovereign risk. This literature has evolved into three main directions. A first branch documents the empirical regularities of fiscal policy in emerging economies and contrasts them to those observed in developed economies. A common theme is that both public spending and taxes are procyclical in the former but much less so in the latter group of countries (see e.g. Gavin and Perotti, 1997; Talvi and Vegh, 2005; Kaminsky, Reinhart, and Vegh, 2005; Frankel, Vegh, and Vuletin, 2013; Vegh and Vuletin, 2015). The explanations advanced to rationalize this pattern include weak political institutions and tax enforcement, incomplete markets and borrowing constraints (see e.g. Tornell and Lane, 1999; Cuadra, Sanchez, and Sapriza, 2010; Ilzetzki, 2011; Bauducco and Caprioli, 2014). Relatedly, a second strand employs models of sovereign default to examine the relationship between fiscal rules or restrictions, bailouts and conditionality imposed by international financial institutions (see e.g. Juessen and Schabert, 2013; Goncalves and Guimaraes, 2015; Hatchondo, Martinez, and Roch, 2015; Fink and Scholl, 2016; Arellano and Bai, 2016). Finally, a third branch integrates fiscal policy into dynamic models of endogenous sovereign default from the perspective of optimal taxation with or without commitment (see e.g. Adam and Grill, 2013; Pouzo and Presno, 2015; Niemann and Pichler, 2016).

Most closely related to our work are the papers by Cuadra, Sanchez, and Sapriza (2010) and Cole and Kehoe (2000). Like us, Cuadra, Sanchez, and Sapriza (2010) consider a small open economy model where not only the repayment of debt but also taxes and public spending are endogenously chosen by the government in a time-consistent fashion. They consider only fundamental default risk and obtain fiscal procyclicality in a framework with exogenous default costs as in Arellano (2008). By contrast, our approach allows us to examine the interaction between fiscal policy and endogenous default costs and accommodates self-fulfilling debt crises, which we show to have important consequences for optimal fiscal policies.⁵ Our detailed account

⁴Section 5.4 below provides further details on this point.

⁵In Gomez-Oliveros Duran, Niemann, and Pichler (2016), we analyze the role of fiscal policy in shaping the

of fiscal policy and default costs also differentiates our work from Cole and Kehoe (2000) from whom we borrow the foundation for rollover risk. While distortionary, taxation in their model is limited to a time-invariant income tax; likewise, productivity is not stochastic and, following default, subject to a permanent exogenous reduction. Instead, our paper integrates flexible and optimally determined fiscal policy into a fully-fledged stochastic environment, which facilitates the quantitative assessment of its properties in normal times and during crisis and transition episodes. This has important implications. Whereas Cole and Kehoe (2000) establish that, generically, the optimal policy is to escape the crisis zone by decumulating debt, our model of optimal fiscal policy rationalizes persistent debt positions within the crisis zone as the generic outcome, which is broken only through rare regime changes following exceptional productivity dynamics. Finally, in a variation of the setup with time-invariant taxation in Cole and Kehoe (2000), Conesa and Kehoe (2015) obtain debt dynamics driven by the government's motive in recessions to 'gamble for resurrection', which may result in a transition into the crisis zone. This is similar to our model, which, however, features empirically plausible productivity dynamics and thus also transitions out of the crisis zone, and in addition details the dynamic pattern of (variable) taxation and spending during these regime switches.

The rest of this paper is organized as follows. Section 2 presents our formal model environment, and Section 3 presents conditions characterizing optimal fiscal policy. Section 4 calibrates the model to data for the Mexican economy. Section 5 presents the results of our quantitative exercise. Section 6 concludes.

2 The model

Our model extends the framework by Mendoza and Yue (2012) by introducing fiscal policy, in the form of public spending and a linear consumption tax, as well as the possibility of self-fulfilling rollover crises (cf. Cole and Kehoe, 2000). We consider a small open economy populated by households, firms and a sovereign government which borrows from foreign lenders. Production is organized in two sectors, a sector f of final goods producers and a sector m of intermediate goods producers. Time is discrete, t = 0, 1, 2, ...

output costs of default in greater detail.

2.1 Private sector

Households. Households choose consumption, c_t , and labor supply, L_t , to maximize a timeseparable utility function,

$$E\sum_{t=0}^{\infty}\beta^{t}u(c_{t}-v(L_{t}),g_{t}),$$
(1)

where g_t denotes valued public expenditure and $\beta \in (0, 1)$ is a discount factor. The period utility function $u(\cdot)$ has standard properties and is separable in its two arguments; the first term complies with the specification in Greenwood, Hercowitz, and Huffman (1988) which removes the wealth effect on labor supply. Households take as given the wage rate w_t , firm profits π_t^f and π_t^m in the two sectors, and the consumption tax τ_t . Since households do not participate in intertemporal asset markets,⁶ their problem consists of maximizing (1) subject to the budget constraint

$$c_t = \frac{1}{1 + \tau_t} \left[w_t L_t + \pi_t^f + \pi_t^m \right].$$
 (2)

The associated optimality condition for labor supply is

$$v'(L_t) = \frac{w_t}{1 + \tau_t}.$$
(3)

Final goods producers. Competitive firms in the f sector combine labor L_t^f , intermediate goods M_t and a time-invariant capital stock k to produce finals goods. The production function is Cobb-Douglas and subject to productivity shocks ε_t ,

$$y_t = \varepsilon_t \left(M\left(m_t^d, m_t^*\right) \right)^{\alpha_M} \left(L_t^f \right)^{\alpha_L} k^{\alpha_k}, \tag{4}$$

⁶The assumption that households cannot borrow directly on international financial markets is widely adopted in the sovereign debt literature. Lahiri, Singh, and Vegh (2007) point out that, in general, only a fraction of households have access to asset markets, and that even in the US the degree of asset market segmentation is remarkably high. Reinhart, Rogoff, and Savastano (2003, 2014) report that, historically, private foreign borrowing amounts to less than 10% of total foreign borrowing in more than two thirds of developing countries. Cuadra, Sanchez, and Sapriza (2010) provide some further discussion. While their baseline model does not allow for private borrowing on international markets, they also study an alternative model with private borrowing and find that their baseline results remain qualitatively unaffected.

where $\alpha_M, \alpha_L, \alpha_k \in (0, 1)$ and $\alpha_M + \alpha_L + \alpha_k = 1$. The mix of intermediate inputs is determined by a CES Armington aggregator combining domestic inputs m_t^d and imported inputs m_t^* ,

$$M_{t} = \left[\lambda \left(m_{t}^{d}\right)^{\mu} + (1-\lambda) \left(m_{t}^{*}\right)^{\mu}\right]^{\frac{1}{\mu}}$$
(5)

with weight $\lambda \in [0, 1)$, implying an elasticity of substitution of $\eta_{m^d, m^*} = \left| \frac{1}{\mu - 1} \right|$. Imported inputs, in turn, are given by a Dixit-Stiglitz aggregator combining a continuum of differentiated varieties $m_j^*, j \in [0, 1],$

$$m_t^* = \left[\int_{j \in [0,1]} \left(m_{jt}^* \right)^{\nu} dj \right]^{\frac{1}{\nu}}, \qquad (6)$$

where $\nu \in (0, 1)$ so that there is a finite elasticity of substitution of $\eta_{m_j^*} = \left|\frac{1}{\nu-1}\right|$ across imported input varieties.

A subset Ω of the imported input varieties, defined by the interval $[0, \theta]$ with $\theta \in (0, 1)$, must be financed in advance via working capital loans κ_t . The timing protocol is such that these within-period loans are contracted after uncertainty about productivity and the government's repayment decision concerning its current debt service has been resolved. As Mendoza and Yue (2012) we assume that the availability of working capital loans to firms conditions on the government's access to international financial markets. Following a sovereign default, working capital loans become unavailable throughout the period of market exclusion. When the government repays, firms can contract loans at the risk-free world interest rate r_t^* ; in this case, their demand for working capital is

$$\frac{\kappa_t}{1+r_t^*} \ge \int_0^\theta p_j^* m_{jt}^* dj,\tag{7}$$

where p_j^* denotes the exogenous, time-invariant price of the imported input variety $j \in [0, 1]$.

Final goods producers choose factor demands to maximize profits, taking w_t , r_t^* , p_j^* and p_t^m ,

the price of domestic inputs, as given. Profits in the final goods sector are given by

$$\pi_t^f = \varepsilon_t \left(M_t \right)^{\alpha_M} \left(L_t^f \right)^{\alpha_L} k^{\alpha_k} - r_t^* \int_0^\theta p_j^* m_{jt}^* dj - \int_0^1 p_j^* m_{jt}^* dj - p_t^m m_t^d - w_t L_t^f \\ = \varepsilon_t \left(M \left(m_t^d, m_t^* \right) \right)^{\alpha_M} \left(L_t^f \right)^{\alpha_L} k^{\alpha_k} - P^*(r_t^*) m_t^* - p_t^m m_t^d - w_t L_t^f,$$
(8)

where $M(m_t^d, m_t^*)$ is given by (5) and $P^*(r_t^*) = \left[\int_0^\theta (p_j^*(1+r_t^*))^{\frac{\nu}{\nu-1}} dj + \int_\theta^1 (p_j^*)^{\frac{\nu}{\nu-1}} dj\right]^{\frac{\nu-1}{\nu}}$ is the price index for imported inputs resulting from CES aggregation. The first-order conditions associated with final goods firms' profit maximization problem are then given by

$$\alpha_M \varepsilon_t \left(M\left(m_t^d, m_t^*\right) \right)^{\alpha_M - \mu} \left(L_t^f \right)^{\alpha_L} k^{\alpha_k} (1 - \lambda) (m_t^*)^{\mu - 1} = P^*(r_t^*), \tag{9}$$

$$\alpha_M \varepsilon_t \left(M \left(m_t^d, m_t^* \right) \right)^{\alpha_M - \mu} \left(L_t^f \right)^{\alpha_L} k^{\alpha_k} \lambda(m_t^d)^{\mu - 1} = p_t^m, \tag{10}$$

$$\alpha_L \varepsilon_t \left(M\left(m_t^d, m_t^*\right) \right)^{\alpha_M} \left(L_t^f \right)^{\alpha_L - 1} k^{\alpha_k} = w_t, \tag{11}$$

and

$$m_{jt}^* = \left(\frac{p_j^*}{P^*(r_t^*)}\right)^{-\frac{1}{1-\nu}} m_t^*, \quad \text{for } j \in [\theta, 1],$$
(12)

$$m_{jt}^* = \left(\frac{p_j^*(1+r_t^*)}{P^*(r_t^*)}\right)^{-\frac{1}{1-\nu}} m_t^* \quad \text{for } j \in [0,\theta].$$
(13)

The expressions in (8), (9), (12) and (13) implicitly assumed the government and firms have access to capital markets. When the country is in default, the relevant price index for imported inputs becomes $P_{aut}^* = \left[\int_{\theta}^{1} (p_j^*)^{\frac{\nu}{\nu-1}} dj\right]^{\frac{\nu-1}{\nu}}$ and the analogs of (12) and (13) are

$$m_{jt}^* = \left(\frac{p_j^*}{P_{aut}^*}\right)^{-\frac{1}{1-\nu}} m_t^*, \quad \text{for } j \in [\theta, 1],$$
(14)

$$m_{jt}^* = 0 \quad \text{for } j \in [0, \theta]. \tag{15}$$

Intermediate goods producers. Competitive firms in the m sector use labor L_t^m to produce intermediate goods according to the production function

$$m_t^d = A(L_t^m)^\gamma,\tag{16}$$

where A > 0 and $\gamma \in [0, 1]$. Taking p_t^m and w_t as given, intermediate firms maximize profits,

$$\pi_t^m = p_t^m A(L_t^m)^\gamma - w_t L_t^m.$$
(17)

The associated optimality condition for labor demand is given by

$$\gamma p_t^m A(L_t^m)^{\gamma-1} = w_t. \tag{18}$$

Finally, notice that labor market clearing requires

$$L_t = L_t^f + L_t^m, (19)$$

and that GDP, the value of the output of final goods net of the costs of imported inputs, is given by

$$gdp_t = y_t - P_t^* m_t^*. (20)$$

2.2 Sovereign government and foreign lenders

The sovereign government implements consumption taxes τ_t , provides public spending g_t and can borrow and lend in international credit markets. Let b_{t+1} denote the amount of debt issued in period t. Financial markets are incomplete because the government can issue only one-period, non-state-contingent discount bonds. The government cannot commit to repay its debt. Each period, conditional on being in good credit standing, the government chooses between honoring its outstanding foreign debt or defaulting on it. The government's inability to commit is the reason why borrowing from international creditors is also subject to self-fulfilling rollover risk: The fear of a future default may prompt lenders not to extend new credit, which in turn may induce an immediate default by the government. There is thus scope for fundamental defaults, driven by the government's willingness to repay, and self-fulfilling rollover crises, driven by a coordination failure between the government and its foreign lenders.

Irrespective of its ultimate source, the consequences of debt repudiation are always the same. Default wipes out the entirety of the government's outstanding debt at the cost of financial autarky, that is, exclusion from international credit markets. When in bad credit standing, the government may regain access to international credit markets in the next period with an exogenous probability ϕ . The government's default decision therefore trades off the direct costs of the resource transfer to international lenders associated with the repayment of the its non-contingent debt against the costs of temporary exclusion from credit markets given by the foregone benefits of consumption smoothing and the output loss in autarky due to the non-availability of working capital to firms.

The government's intertemporal problem can be expressed in recursive form. The fundamental state variables are the government's inherited bond position b and the current productivity level ε . In order to examine the implications of rollover risk, we follow Chatterjee and Eyigungor (2012) and consider the following static coordination game played at the beginning of each period between the sovereign government and foreign lenders.⁷ The government faces some maturing debt b and, conditional on redeeming its current liabilities, seeks to issue new bonds. Table 1 details the relevant payoff matrix, whereby columns correspond to the sovereign government's strategies, and rows correspond to the lenders' strategies. If lenders extend new

Table 1: Payoff matrix for static coordination game

	repay	default
rollover	$0, \mathcal{V}^{nd}(b, \varepsilon)$	$-r^*/(1+r^*)\Delta, \mathcal{V}^d(\varepsilon)$
crisis	$0, \mathcal{V}^c(b, \varepsilon)$	$0, \mathcal{V}^d(\varepsilon)$

credit (rollover) and the government repays its maturing debt (repay), the former earn a net return of zero (that is, in expectation lenders earn the return r^* which is also the opportunity cost of their funds), and the latter receives the payoff under repayment and new borrowing, denoted by $\mathcal{V}^{nd}(b,\varepsilon)$. If lenders extend new credit and the government defaults (default), we assume that the new borrowing is returned to the lenders without it earning any interest; the (discounted) loss of interest earnings then is $r^*/(1 + r^*)\Delta$, where Δ is the amount of new lending. If lenders fail to provide new credit (crisis), their payoff is zero irrespective of the government's behavior. In this case, if the government repays but cannot borrow, it receives $\mathcal{V}^c(b,\varepsilon) \leq \mathcal{V}^{nd}(b,\varepsilon)$.⁸ And if the government defaults, its payoff does not depend on the level

⁷For simplicity, and as in Chatterjee and Eyigungor (2012), we abstract from the coordination problem between multiple lenders and assume they act in a coordinated fashion. In Cole and Kehoe (2000), the game is played between the sovereign and many lenders acting independently.

⁸The weak inequality holds because the policy when fresh borrowing is allowed can always replicate the

of maturing debt and is given by $\mathcal{V}^d(\varepsilon)$.

We assume indifference on the side of the government or lenders is always resolved in favor of repayment and extending new credit, respectively. Depending on the value of $\mathcal{V}^d(\varepsilon)$, the game then has the following set of Nash equilibria. If $\mathcal{V}^d(\varepsilon) \leq \mathcal{V}^c(b,\varepsilon) \leq \mathcal{V}^{nd}(b,\varepsilon)$, the unique equilibrium is (rollover, repay); if $\mathcal{V}^c(b,\varepsilon) \leq \mathcal{V}^{nd}(b,\varepsilon) < \mathcal{V}^d(\varepsilon)$, the unique equilibrium is (crisis, default); and if $\mathcal{V}^c(b,\varepsilon) < \mathcal{V}^d(\varepsilon) \leq \mathcal{V}^{nd}(b,\varepsilon)$, both (rollover, repay) and (crisis, default) are equilibria. In the last case, we assume that the equilibrium is selected depending on the realization of a sunspot variable, denoted ξ . If $\xi = 0$, the (rollover, repay)equilibrium is selected, and if $\xi = 1$, the (crisis, default) equilibrium is selected. The latter case corresponds to a self-fulfilling rollover crisis where lenders refuse to lend because they believe that the sovereign will default, and the sovereign defaults because it believes that the lenders will refuse to lend. We assume the sunspot variable ξ is i.i.d. and takes value one with probability p.

Let $\mathcal{V}(b, \varepsilon, \xi)$ denote the value function of the government, which depends on the fundamental states (b, ε) and the sunspot variable ξ . If the government has access to credit and does not default, it can issue new debt and finance expenditures subject to the following constraint,

$$g = \tau c + q(b', \varepsilon)b' - b, \tag{21}$$

where $q(b', \varepsilon)$ is the bond pricing function, which is taken as given by the government. When implementing its policy, the government also needs to take into account the private sector's response given by the set of optimality conditions

$$\mathcal{E}^{nd} = \{(2), (3), (4), (7), (9), (10), (11), (12), (13), (16), (18), (19)\}.$$

Thus, the government problem conditional on repayment is

$$\mathcal{V}^{nd}(b,\varepsilon) = \max_{\tau,g,b'} \left\{ u(c-v(L),g) + \beta E\left[(1-p)\mathcal{V}(b',\varepsilon',0) + p\mathcal{V}(b',\varepsilon',1) \right] \right\}$$
(22)

subject to (21) and \mathcal{E}^{nd} . By contrast, if the government defaults, it is temporarily excluded policy without borrowing, or even do better.

from international credit markets and faces the constraint

$$g = \tau c, \tag{23}$$

and the private-sector optimality conditions \mathcal{E}^d , where, relative to \mathcal{E}^{nd} , $\kappa = 0$, the import price index is P_{aut}^* and (12) and (13) are replaced by (14) and (15). The problem of a defaulting government then is

$$\mathcal{V}^{d}(\varepsilon) = \max_{\tau,g} \left\{ u(c - v(L), g) + \beta E \left[(1 - \phi) \mathcal{V}^{d}(\varepsilon') + \phi \left[(1 - p) \mathcal{V}(0, \varepsilon', 0) + p \mathcal{V}(0, \varepsilon', 1) \right] \right\}$$
(24)

subject to (23) and \mathcal{E}^d . And similarly, the value function when international lenders refuse to extend new credit is given by

$$\mathcal{V}^{c}(b,\varepsilon) = \max_{\tau,g} \left\{ u(c-v(L),g) + \beta E\left[(1-p)\mathcal{V}(0,\varepsilon',0) + p\mathcal{V}(0,\varepsilon',1)\right] \right\}$$
(25)

subject to $g = \tau c - b$ and $\mathcal{E}^{d,9}$ Given $\mathcal{V}^{nd}(b,\varepsilon)$, $\mathcal{V}^{d}(\varepsilon)$ and $\mathcal{V}^{c}(b,\varepsilon)$, the government's value function is determined as

$$\mathcal{V}(b,\varepsilon,\xi) = \begin{cases}
\mathcal{V}^{nd}(b,\varepsilon), & \text{if } \mathcal{V}^{d}(\varepsilon) \leq \mathcal{V}^{c}(b,\varepsilon) \text{ and } \xi \in \{0,1\}, \\
\mathcal{V}^{d}(\varepsilon), & \text{if } \mathcal{V}^{nd}(b,\varepsilon) < \mathcal{V}^{d}(\varepsilon) \text{ and } \xi \in \{0,1\}, \\
\mathcal{V}^{nd}(b,\varepsilon), & \text{if } \mathcal{V}^{c}(b,\varepsilon) < \mathcal{V}^{d}(\varepsilon) \leq \mathcal{V}^{nd}(b,\varepsilon) \text{ and } \xi = 0, \\
\mathcal{V}^{d}(\varepsilon), & \text{if } \mathcal{V}^{c}(b,\varepsilon) < \mathcal{V}^{d}(\varepsilon) \leq \mathcal{V}^{nd}(b,\varepsilon) \text{ and } \xi = 1.
\end{cases}$$
(26)

The optimal default policy under rollover risk can be characterized with reference to (26) as

$$\mathcal{D}(b,\varepsilon,\xi) = \begin{cases} 0, & \text{if } \mathcal{V}(b,\varepsilon,\xi) = \mathcal{V}^{nd}(b,\varepsilon), \\ 1, & \text{if } \mathcal{V}(b,\varepsilon,\xi) = \mathcal{V}^{d}(\varepsilon). \end{cases}$$
(27)

This determines a default set, the set of productivity realizations such that, given the sunspot

⁹Implicit in this formulation is the assumption that final goods firms do not have access to working capital loans when lenders refuse to buy government bonds. This parallels our earlier assumptions about the consequences of sovereign default.

and current debt, default is optimal,

$$\Gamma(b,\xi) = \{\varepsilon : \mathcal{D}(b,\varepsilon,\xi) = 1\}.$$
(28)

Moreover, given current productivity ε and some debt policy b', the probability of default in the next period can be inferred from the default set and the transition process $z(\varepsilon'|\varepsilon)$ for productivity as

$$\rho(b',\varepsilon) = (1-p) \int_{\Gamma(b',0)} dz(\varepsilon'|\varepsilon) + p \int_{\Gamma(b',1)} dz(\varepsilon'|\varepsilon).$$
(29)

Bond prices are determined by international lenders, who are risk-neutral and have complete information. Facing an opportunity cost of funds equal to r^* , they invest in one-period sovereign bonds and in within-period private working capital loans. Competition implies that lenders expect zero profits and that the returns on sovereign debt and the world's risk-free asset are fully arbitraged, that is,

$$q(b',\varepsilon) = \frac{1-\rho(b',\varepsilon)}{1+r^*}.$$
(30)

Accordingly, the equilibrium bond price $q(b', \varepsilon)$ reflects the risk of sovereign default.

Equilibrium. In equilibrium, private-sector allocations are optimal, given the policies implemented by the government; the government's debt, default and fiscal policy are optimal subject to the relevant implementability constraints and the bond pricing function $q(b', \varepsilon)$; and foreign lenders are optimizing.

3 Optimal fiscal policy

We now examine the properties of optimal fiscal policy. A first result is that the monotonicity of the optimal (discretionary) default policy, which is familiar from models following Eaton and Gersovitz (1981), is preserved also in the presence of optimal fiscal policy.

Proposition 1. Given a productivity shock ε and a pair of debt positions b and \tilde{b} such that $b > \tilde{b} \ge 0$, if default is optimal for \tilde{b} , then it is optimal also for b. That is, $\Gamma(\tilde{b}, \xi) \subseteq \Gamma(b, \xi)$.

Since bond prices compensate for default risk, it follows that, for given current productivity, they must also be monotonic in the amount of debt issued. Indeed, the government's lack of commitment matters only if bond prices react to debt. This is because both taxation and public spending affect only the static equilibrium conditions. The following proposition characterizes the government's optimal tax and spending policy.

Proposition 2. The government's optimal tax policy satisfies

$$u_c \left\{ -\frac{\partial \frac{1}{1+\tau}}{\partial \tau} g dp \right\} = u_g \left\{ -\frac{\partial \frac{1}{1+\tau}}{\partial \tau} g dp + \frac{\tau}{1+\tau} \frac{\partial g dp}{\partial \tau} \right\},$$

which implies underprovision of public spending, $u_c < u_g$.

This condition has an interpretation in terms of marginal benefits and marginal costs of changing the tax rate. Variations in the government's tax policy are then seen to have two effects: a direct reallocation effect $\left(-\frac{\partial \frac{1}{1+\tau}}{\partial \tau}gdp > 0\right)$, and a budgetary effect $\left(\frac{\tau}{1+\tau}\frac{\partial gdp}{\partial \tau} < 0\right)$. In detail, for given GDP, an increase in the tax rate allows to reallocate resources from private to public consumption. However, this causes tax distortions which work to reduce GDP, the relevant tax base for the consumption tax, and thus has negative implications for the government's budget. In conjunction, these effects imply that the optimal fiscal policy limits distortions by keeping public expenditure below its first-best level.

To build intuition, it is also useful to consider the generalized Euler equation characterizing the government's optimal debt policy.¹⁰

Proposition 3. The government's optimal debt policy satisfies

$$u_g\left\{q+\frac{\partial q}{\partial b'}b'\right\} = \beta E_{\xi'}E_{\varepsilon'\notin\Gamma(b',\xi')}u'_g.$$

An immediate implication of the generalized Euler equation is that increasing debt can only be optimal for the government if this generates additional resources, that is, $q + \frac{\partial q}{\partial b'}b' > 0$. So the optimal debt policy can never be subject to a debt Laffer curve. More generally, the generalized

¹⁰Technically, this presumes differentiability of the bond pricing function $q(b', \varepsilon)$ and the value function $\mathcal{V}^{nd}(b, \varepsilon)$. Notice therefore that the generalized Euler equation is presented merely to illustrate the intertemporal policy trade-off facing the government. We do not claim or prove differentiability, and also our numerical approach and quantitative results do not rest on it.

Euler equation balances today's marginal benefit of additional borrowing against the discounted marginal cost of higher debt tomorrow. The marginal benefit values with the marginal utility of public spending the additional resources available to the government from issuing an extra unit of debt. The marginal cost of higher inherited liabilities is to reduce the resources available for future public spending. Notice, however, that the possibility of a future default implies that the government takes this cost into account only for states which actually induce repayment.¹¹ Since the government will exercise its default option exactly in times when repayment would be associated with an excessively high marginal utility from public spending, this conditioning implies $E_{\xi'}E_{\varepsilon' \not\in \Gamma(b',\xi')}u'_g \leq E_{\xi'}E_{\varepsilon'}u'_{\bar{g}}$, where $E_{\xi'}E_{\varepsilon'}u'_{\bar{g}}$ is the marginal utility under the suboptimal policy dictating repayment in all states (ε, ξ). How the government resolves the intertemporal trade-off underlying its debt policy therefore depends on the strength of its effective fronloading motive, as captured by its discount factor β and the reduction in the expected future marginal utility via the option value of default.

4 Calibration

We now study the model's quantitative implications in a calibrated environment. We target data for Mexico for two reasons. First, given our emphasis on fiscal policy, we can rely on time series data for public finances, whose availability and quality is better in Mexico compared to many other emerging economies including Argentina. Second, the Mexican debt crisis of 1994-1995 is widely interpreted as driven by self-fulfilling dynamics of the sort considered in our model. Notice also that, at that time, the maturity structure of sovereign debt was very low, with the majority of bonds having a maturity of just 91 days and overall maturity averaging at barely 200 days (Cole and Kehoe, 1996). Our quarterly model with one-period debt is thus a good approximation.

Similar to Cuadra, Sanchez, and Sapriza (2010), the period utility function in (1) is assumed

¹¹Key to this result is the fact that the level of maturing debt becomes irrelevant in the event of default. It therefore also does not matter whether a default occurs for fundamental or self-fulfilling reasons.

to take the following form,

$$u(c-v(L),g) = \pi \left(\frac{(c-\frac{L^{\omega}}{\omega})^{1-\sigma}}{1-\sigma}\right) + (1-\pi)\left(\frac{g^{1-\sigma}}{1-\sigma}\right).$$
(31)

Accordingly, the contributions of public expenditures and the consumption-leisure composite to utility are subject to the same intertemporal elasticity of substitution $\frac{1}{\sigma}$ and are aggregated with relative weights $(1-\pi)$ and π . Labor supply is characterized by a constant Frisch elasticity of $\frac{1}{\omega-1}$. The benchmark parameterization for our quarterly model is summarized in Table 2. Parameters above the line are calibrated. The curvature parameters are set to $\sigma = 2$ and $\omega = 1.455$, corresponding to a Frisch elasticity of $\frac{1}{\omega-1} = 2.2$. These are standard values in quantitative studies of sovereign default and international real business cycles (Mendoza, 1991; Neumeyer and Perri, 2005; Cuadra, Sanchez, and Sapriza, 2010; Mendoza and Yue, 2012). Likewise, the quarterly risk-free interest rate r^* is set to 1%. Given the other parameter values assigned in (31), the preference weight is obtained as $\pi = 0.9$, which generates a ratio of public to private consumption of 20%, the mean value observed in Mexico (1980-2009). The probability of reentry after default is set at $\phi = 0.083$; the implied average exclusion period of three years is consistent with relevant empirial estimates (Dias and Richmond, 2009; Gelos, Sahay, and Sandleris, 2011; Cruces and Trebesch, 2013).

Our calibration of the parameters relating to the aggregation of intermediate inputs in (5) and (6) follows Mendoza and Yue (2012) who recur on Mexican data (1988-2004) and infer $\lambda = 0.62$ and $\mu = 0.65$ from optimality conditions (9) and (10).¹² There is thus a small bias in favor of domestic relative to imported inputs, and the elasticity of substitution between them is $\eta_{m^d,m^*} = \frac{1}{1-\mu} = 2.86$. The parameter ν is pinned down by the elasticity of substitution across imported varieties, which – building on evidence reported in Gopinath and Neiman (2014) – is set at $\eta_{m_j^*} = \frac{1}{1-\nu} = 2.44$ as in Mendoza and Yue (2012); this implies $\nu = 0.59$. Finally, the target for θ is the share of working capital financing in GDP. As Schmitt-Grohe and Uribe (2007) and Mendoza and Yue (2012), we proxy working capital by the fraction of M1 held by firms, whereby we rely on an estimate for the US showing that firms hold about two-thirds of M1. On the basis of this strategy, our estimate for the importance of working capital results

¹²These two values allow their model to match the average ratios in Mexican data of imported to domestic inputs at current and constant prices, which are 18% and 15.7%, respectively.

Parameter		Value	Source / Target statistics
Inverse Frisch elasticity	ω	1.455	Standard value
Inverse of intertemporal elasticity of consumption	σ	2	Standard value
Risk-free rate	r^*	0.01	Standard value
Reentry probability	ϕ	0.083	Standard value
Armington weight	$\stackrel{'}{\lambda}$	0.62	Mendoza and Yue (2012)
of domestic inputs			
Armington curvature	μ	0.65	Mendoza and Yue (2012)
Dixit-Stiglitz curvature	ν	0.59	Mendoza and Yue (2012)
Working capital to GDP ratio	θ	0.7	Mendoza and Yue (2012)
International goods share in gross output of final goods	α_M	0.43	Mendoza and Yue (2012)
Labor share in gross output of final goods	α_L	0.26	Labor income share in GDP (0.46)
Labor share in production of intermediate goods	γ	0.46	Labor income share in GDP (0.46)
Private cons. weight in utility	π	0.9	Public-private consumption ratio (0.2)
Intermed. sector productivity	A	0.18	Quarterly output drop in default (5%)
Autocorrelation of TFP shock	ρ_{ε}	0.98	GDP autocorrelation (0.84)
Standard dev. of TFP shock	σ_{ε}	0.01	GDP standard dev. (0.024)
Discount factor	β	0.94	Priv. consumption standard dev. (0.029)
Sunspot probability	p	0.01	Average annual spread (4.27%)

 Table 2: Parameter values

-

in $\theta = 0.7$. The share of intermediate goods in gross output is set to $\alpha_M = 0.43$ as in Mendoza and Yue (2012). We assume identical labor shares in the final (f) and intermediate (m) goods sectors. We set $\gamma = 0.46$, which, according to OECD's Unit Labour Costs Annual Indicators data, is the average labor income share in Mexico during the period 1980-2009. A labor income share in value added of the f sector of $\frac{\alpha_L}{1-\alpha_M} = \gamma = 0.46$ then implies $\alpha_L = 0.26$. Under constant returns to scale in the f sector, we finally have $\alpha_k = 1 - \alpha_M - \alpha_L = 0.31$.

Parameters below the line are set with SMM, similar to Mendoza and Yue (2012). Productivity shocks in final goods production are assumed to follow an AR(1) process,

$$\ln \varepsilon' = \rho_{\varepsilon} \ln \varepsilon + \epsilon', \tag{32}$$

with $E(\epsilon) = 0$ and $E(\epsilon^2) = \sigma_{\epsilon}$. This productivity process is approximated by a discrete firstorder Markov chain with 25 values using the procedure in Tauchen (1986). The parameters ρ_{ϵ} and σ_{ϵ} are then chosen to target the autocorrelation and standard deviation of quarterly GDP in Mexico (1980-2009, H-P filtered, OECD Quarterly National Accounts). The data moments are 0.84 and 2.4%, and the resulting parameter values are $\rho_{\epsilon} = 0.98$ and $\sigma_{\epsilon} = 0.01$. The remaining parameters A, β and p are chosen to target the decline in output at default, which is approximately 5% in the data; the standard deviation of HP-detrended private consumption, which is approximately 3%; and the average annualized interest rate spread, which equals 4.27% in the EMBI data provided by Cuadra, Sanchez, and Sapriza (2010). Given these targets, the SMM procedure yields A = 0.18, $\beta = 0.94$ and p = 0.01.

Some comments are in order. While our calibration of the discount factor at $\beta = 0.94$ is quite high in comparison to much of the sovereign default literature (e.g. Mendoza and Yue, 2012 calibrate $\beta = 0.88$), it remains low relative to the values typically assigned in quarterly business cycle models. This has two consequences for the simulation of our model and its quantitative interpretation. On the one hand, since the sovereign government is relatively patient, its frontloading motive is weaker than in most comparable papers concerned with sovereign default. Hence, the accumulation of debt remains moderate and the model generates relatively few fundamental default episodes. In order to match the empirically observed interest rate spreads, rollover risk driven by sunspots must fill the gap. We see this as an interesting and empirically relevant feature of our model, which has important implications for the optimal conduct of fiscal policy. On the other hand, annual discount rates implied by β are still very high at around 28%. This seems hard to justify simply on the grounds of politico-economic distortions. The meaningful analysis of the trade-offs shaping optimal fiscal policy hence arguably calls for consideration of environments with a less dominant frontloading motive. Therefore, we complement our baseline scenario with $\beta = 0.94$ by alternative parameterizations with higher values for the discount factor at $\beta = 0.96$ and $\beta = 0.98$.

5 Results

In this section we present the results of our quantitative analysis. We start by establishing the existence of a *crisis zone*, that is, a region of the state space where self-fulfilling debt crises are possible, and examine the implications for the pricing of sovereign debt. We then move on to discuss the business cycle implications of our model and show that optimal fiscal policy is generally procyclical. Finally, we show that there are infrequent episodes of severely countercyclical fiscal policy, which occur when the economy moves into our out of the crisis zone, and we examine in detail the economic conditions underlying these *regime shift* events.

5.1 Crisis zone and bond prices

As in Cole and Kehoe (2000), the state space (b, ε) in our model can be partitioned into three zones: the *safe zone* where the government always prefers to honor its maturing debt, the *default zone* where the government always prefers default, and the *crisis zone* where there is scope for self-fulfilling debt crises. Figure 1 visualizes the different zones for the case when $\beta = 0.94$. The crisis threshold moves from about 4.5% to about 5.3% of annual GDP when TFP ranges within two standard deviations of its mean. When debt exceeds this threshold, self-fulfilling sovereign debt crises may occur depending on the realization of the sunspot variable ξ . The default threshold is in the order of magnitude of 9% of annual GDP. Interestingly, the default threshold expressed in terms of a debt-to-GDP ratio is (slightly) non-monotonic in TFP. This is because in low TFP states the direct effect of increasing productivity on GDP is stronger than its effect on sustainable debt. Maybe more importantly, Figure 1 illustrates that the crisis



Figure 1: Crisis and default zones under optimal fiscal policy ($\beta = 0.94$)

zone in our model is pervasive, so that rollover risk is a concern over a large part of the state space. This has important implications for the behavior of bond prices. Figure 2 plots the bond pricing function (30) against the level of debt when $\varepsilon = 1$. As seen, there are two regions where



Figure 2: Bond pricing function

bond prices display a strong sensitivity to the amount of debt issued. The first drop in bond prices occurs when the economy approaches the crisis zone and default risk rises from zero to p. The second drop arises due to the surge in default risk when the country approaches the default zone.¹³ This begs the question of how rollover risk is addressed by optimal fiscal policy. Underlying the properties of optimal fiscal policy is the trade-off between the government's motives to frontload and to smooth consumption. The answer to the above question therefore critically depends on the relative importance of the government's frontloading motive, which is parameterized by the discount factor β .

5.2 Fiscal procyclicality

Table 3 presents business cycle statistics generated from simulations of our model and compares them to quarterly data from Mexico (1980-2009). For the benchmark model with $\beta = 0.94$, the average level of debt over GDP is around 8%, which is well within the crisis zone. Table 3 shows further that our benchmark model reproduces several salient features of the Mexican business cycle: both private and public consumption are more volatile than GDP, whereas taxes, interest rate spreads and the trade balance are less volatile. Note that the relative volatility of spreads is significantly lower than in the data. This is because default risk varies very little over the business cycle. In particular, fundamental default risk is very low, while rollover risk does not vary with the level of debt within the crisis zone.¹⁴

The pattern of correlations implied by our model shows that default risk is countercyclical. This induces a procyclical debt policy, which manifests itself in a countercyclical trade balance. Despite this, spreads are countercyclical and positively correlated with the trade balance. Key to this correlation pattern is the convexity of the default costs in the underlying productivity state. Following Mendoza and Yue (2012), our model generates this structure endogenously. Turning to the cyclical properties of the other fiscal instruments, we see that public spending displays a tight positive correlation with GDP, whereas tax rates display a negative correlation. This pattern is robust to changes in the discount factor β . Fiscal policy in our model is thus

¹³Similar two-step bond pricing functions can arise also in the context of default models with political risk (Hatchondo, Martinez, and Sapriza, 2009; Scholl, 2017), when the hazard of turnover to a more default-prone government induces bond prices akin to those depicted in Figure 2.

¹⁴Recall that our calibration implies that fundamental default risk contributes only 0.27 percentage points to the average annual interest rate spread of 4.27%. Nevertheless, it is the threat of fundamental default as opposed to rollover risk that acts as the key constraint on fiscal policy; see Appendix A.2.

	(I)	(II)	(III)	$\overline{(IV)}$	
Statistic	Data	$\beta=0.94$	$\beta=0.96$	$\beta=0.98$	
	1	1 1	f ann		
Standard deviation relative	to stanad	ara aeviatio	on of GDP		
C	1.20	1.28	1.30	1.69	
G	1.08	1.39	1.44	2.01	
au	0.24	0.54	0.67	1.29	
R_s	0.82	0.04	0.09	0.01	
TB	0.65	0.47	0.35	0.22	
Correlation with GDP					
C	0.93	0.95	0.92	0.86	
G	0.38	0.92	0.87	0.81	
au	-0.33	-0.46	-0.35	-0.48	
R_s	-0.64	-0.21	-0.23	-0.42	
TB	-0.68	-0.47	-0.33	-0.31	
D	-0.11	-0.13	-0.13	-0.17	
Correlation with spread					
C	-0.56	-0.22	-0.21	-0.43	
G	-0.16	-0.23	-0.20	-0.45	
TB	0.65	0.18	0.03	0.32	
Mean debt-to-annual GDP	29%	7.6%	6.8%	4.8%	
Mean annual spread	4.27%	4.27%	3.25%	0.02%	
Time spent in autarky	-	10.71%	8.28%	0.04%	

Table 3: Business cycle statistics

Note: The statistics reported are computed as averages over N = 2000 simulations of length T = 500, with the first 100 observations truncated. The simulated series are logged and filtered. The empirical measure for taxes is based on data for 1980-2007 and taken from Cuadra, Sanchez, and Sapriza (2010).

generally procyclical.¹⁵

A closer look reveals, however, that the presence of rollover risk gives rise to infrequent episodes of strongly countercyclical fiscal policy. Figure 3 shows the debt positions generated by our model specifications, together with the relevant crisis and default thresholds. The first panel considers the case when $\beta = 0.94$ so that the government's frontloading motive is very pronounced. As seen, the simulated debt positions are generally close to the default threshold. There is a tight comovement between the two time series, which illustrates again the procyclical nature of both the government's borrowing and its incentives to repay. However,

¹⁵In this respect, our findings confirm the results in Cuadra, Sanchez, and Sapriza (2010), though for an environment with rollover risk and endogenous default costs. Notice also that variations in β actually have a non-monotonic effect on the degree of procyclicality in taxation; this can be explained through the changing incidence of market exclusion.



Figure 3: Debt, crisis threshold and default threshold for varying discount factor

in rare instances, self-fulfilling rollover crises or strong contractions in productivity and GDP trigger episodes of sovereign default, which lead to periods of market exclusion. But once the government regains access to international capital markets, debt positions are fast to reach the vicinity of the default threshold again. Moreover, under the optimal fiscal policy, there is no attempt to escape the crisis zone.

By contrast, the last panel examines the situation for $\beta = 0.98$, implying a relatively weak frontloading motive. Debt positions are again procyclical but now remain below the crisis zone. This is because default risk and bond prices would jump discontinuously if the country were to enter the crisis zone. The government thus limits its borrowing and eliminates default risk almost completely.¹⁶

The most interesting dynamics arise in the intermediate case when $\beta = 0.96$. Over the simulation, debt positions actually range from the safe zone to the crisis zone and even the default zone. Hence, we observe sovereign defaults as well as endogenous regime switches where the government enters or escapes the crisis zone. Notably, though, under an intermediate degree of impatience, debt displays considerable persistence both in the safe zone and in the crisis zone: Below the crisis threshold, the government will *normally* limit borrowing in order to prevent rollover risk – similar to panel (c). But above the crisis threshold, it will *normally* not be willing to shoulder the cost of decumulating debt; instead, since the degree of rollover risk does not vary within the crisis zone, debt levels converge close to the default threshold – similar to panel (a). Occasionally, however, the government finds it worthwhile to engineer a transition into or out of the crisis zone. The next section looks at the economic conditions triggering these transition events and the associated dynamics of fiscal policy in greater detail.

5.3 Countercyclical fiscal expansions and contractions

It is instructive to assess the transition dynamics through the lense of the optimality condition presented in Proposition 3. Presuming differentiability (again, this is just done to ease interpretation) and defining $\epsilon_q(b') = \frac{\partial q}{\partial b'} \frac{b'}{q} \leq 0$ as the elasticity of bond prices with respect to debt

¹⁶Some minimal default risk remains due to the small hazard of a transition into the crisis zone.

issuance, this condition can be rewritten as

$$u_g \left\{ 1 + \epsilon_q(b') \right\} = \frac{\beta}{q} E_{\xi'} E_{\varepsilon' \notin \Gamma(b',\xi')} u'_g.$$
(33)

Accordingly, the intertemporal smoothing of the marginal utility from public spending is complicated by three potential factors. On the right-hand side of (33), both the government's relative impatience, $\beta/q \approx \beta(1+r^*) < 1$, and the insurance value implied by its future default option, $E_{\xi'}E_{\varepsilon'\notin\Gamma(b',\xi')}u'_g \leq E_{\xi'}E_{\varepsilon'}u'_{\tilde{g}}$, provide incentives for higher current spending (lower u_g). Such policy comes at the cost of increased debt and consequently lower bond prices, $\epsilon_q(b') \leq 0$. Starting in the safe zone, these forces can dominate the government's precautionary motive and instead make it worthwhile to risk a transition into the crisis zone. Different from the default frontier, bond prices at the crisis threshold are locally very sensitive to the level of debt, but then flatten out and do not fall to zero (see Figure 2). This is because the crisis risk is small (p = 0.01) and invariant to the level of debt. As a consequence, the effect on bond prices remains limited so that the joint effect of impatience and the option value of default can render a probabilistic (namely after a sufficiently bad productivity shock) transition into the crisis zone optimal. A similar logic applies to transitions out of the crisis zone, although the government's relative impatience makes them less likely. In this light, Figures 4 and 5 illustrate the model's prescriptions for macroeconomic dynamics, and in particular fiscal policy, during such transition episodes. They present simulated time-series data for an event window of eight quarters before and after transition events recorded in the model simulations.¹⁷

Figure 4 shows that transitions *into* the crisis zone are driven by a sequence of bad productivity draws. In response to that, and despite fiscal effort in the form of increased taxation and cuts in public spending to contain an excessive accumulation of debt, the debt-to-GDP ratio increases and the crisis threshold declines, so that the economy approaches the crisis zone. At some point (normalized as t = 0), staying out of the crisis zone becomes too costly as formalized by (33). The government then terminates its procyclical policy and implements a large debt-financed fiscal expansion, reducing taxes and increasing public spending significantly. This policy has a positive effect on contemporaneous output, but comes at the cost of pushing the

¹⁷Since endogenous transitions are most likely to occur when the government's frontloading motive is intermediate, the Figures examine the case when $\beta = 0.96$.



Figure 4: Fiscal policy when entering the crisis zone ($\beta = 0.96$)

economy into the crisis zone as reflected by depressed bond prices. As productivity recovers, the economy moves to its new normal within the crisis zone, with a significantly higher level of debt and, accordingly, a lower level of government spending and higher tax rates compared to the initial situation.

Figure 5 examines fiscal policy when the economy *leaves* the crisis zone. The vertical dashed lines indicate two points in time, the actual transition event from the crisis zone to the safe zone (normalized as t = 0) and the reversal in debt-to-GDP positions when this transition is initiated (on average a year earlier in t = -4). During booms, public spending and the level of debt initially rise in a procyclical fashion. Given its frontloading motive and the procyclical evolution of the default frontier, the government makes use of its fiscal space and issues additional debt. Eventually, however, the accumulated debt burden is so high that the implied decrease in expected future consumption is sufficiently strong to induce a reversal in the government's trade-off between frontloading and smoothing consumption. The government then finds it optimal to implement a severe fiscal contraction in order to escape the crisis zone and to create fiscal space for the future. The associated fiscal tightening involves both a substantial hike in tax rates and cuts in public spending. This induces a fast consolidation of



Figure 5: Fiscal policy when leaving the crisis zone ($\beta = 0.96$)

debt, which is driven below the crisis threshold within an average transition period of only four quarters.

Another important take-away is that consolidations to leave the crisis zone occur only during exceptional boom episodes, when productivity and output are significantly above trend. Indeed, the debt reversal at t = -4 occurs when TFP is almost two standard deviations above average. This is remarkable insofar as movements into the crisis zone happen already during relatively mild recessions (cf. Figure 4, where the regime switch occurs when TFP is just about half a standard deviation below average). There is thus a fundamental asymmetry, which distinguishes transition events into and out of the crisis zone. This asymmetry can be explained with reference to condition (33), whose right-hand-side contains two factors implying a tendency towards debt accumulation rather than decumulation: the government's relative impatience and the discounting of the expected future marginal utility via the option value of default.

5.4 Empirical implications

We conclude this section by pointing out that the normative implications regarding optimal fiscal policy during transition episodes, that is, (i) the existence of large, fast-paced and countercyclical swings in government debt, and (ii) the asymmetry in the productivity trigger underlying these events, are supported by empirical evidence. Abbas, Belhocine, El-Ganainy, and Horton (2011) examine historical debt dynamics in advanced and emerging economies (1870-2007) and decompose 128 identified episodes of large debt increases and decreases into their respective budgetary determinants.¹⁸ They document that strong growth is a consistent feature of most pronounced debt consolidations; (peacetime) debt buildups tend to be driven by weak growth or recessions, but in comparison to debt reductions the picture is somewhat less clearcut. Another takeaway is that the documented *large* debt swings appear hard to rationalize as the cyclical response to underlying productivity shocks. Instead, our model, where – conditional on being in the crisis zone – rollover risk is not tied to fundamentals and thus constant, offers a plausible explanation: When fiscal policy is subject to a frontloading motive, the optimal policy exploits the fact that bond prices remain locally invariant to the level of liabilities and brings debt close to the default frontier. And conversely, successful fiscal consolidations must bring down the stock of debt sufficiently to escape the crisis zone.

6 Conclusions

This paper integrates rollover risk into a model of sovereign debt sustained by endogenous default costs. We use this framework to study how sovereign risk – both fundamental and self-fulfilling – shapes the government's optimal debt policy and the cyclical behavior of taxes and public spending. Central to the fiscal policy problem is the trade-off between the government's motives to frontload and to smooth consumption. Optimal taxes and spending are generally procyclical, but the incidence of rollover risk gives rise to infrequent episodes of severely countercyclical fiscal activity. These regime changes occur endogenously when the economy enters or

 $^{^{18}}$ Even though Abbas, Belhocine, El-Ganainy, and Horton (2011) restrict their decomposition exercise to advanced economies, their data shows that large debt increases and decreases are common also in emerging economies and low-income countries. Notice also that debt reductions are engineered through default only in a small number (7) of the recorded cases (68).

leaves the crisis zone. Transitions into the crisis zone occur following relatively mild recessions that are sustained enough to make the government adopt expansionary measures in the form of tax cuts, an increase in spending and the issuance of public debt, although this increases future borrowing costs. By contrast, transitions out of the crisis zone happen only during exceptional boom periods, which are exploited by the government as an opportunity to decumulate debt in order to eliminate rollover risk and reduce future borrowing costs. In normal times, however, it is too costly for the government to escape the crisis zone, so that debt displays substantial persistence and is issued in a procyclical fashion.

References

- ABBAS, S. M. A., N. BELHOCINE, A. EL-GANAINY, AND M. HORTON (2011): "Historical Patterns and Dynamics of Public DebtEvidence From a New Database," *IMF Economic Review*, 59(4), 717–742.
- ADAM, K., AND M. GRILL (2013): "Optimal sovereign default," Discussion Papers 09/2013, Deutsche Bundesbank, Research Centre.
- ARELLANO, C. (2008): "Default Risk and Income Fluctuations in Emerging Economies," American Economic Review, 98(3), 690–712.
- ARELLANO, C., AND Y. BAI (2016): "Fiscal Austerity during Debt Crises," Staff Report 525, Federal Reserve Bank of Minneapolis.
- BAUDUCCO, S., AND F. CAPRIOLI (2014): "Optimal fiscal policy in a small open economy with limited commitment," *Journal of International Economics*, 93(2), 302–315.
- BORN, B., G. MUELLER, AND J. PFEIFER (2015): "Does austerity pay off?," CEPR Discussion Papers 10425, C.E.P.R. Discussion Papers.
- CALVO, G. A. (1988): "Servicing the Public Debt: The Role of Expectations," American Economic Review, 78(4), 647–61.
- CHATTERJEE, S., AND B. EYIGUNGOR (2012): "Maturity, Indebtedness, and Default Risk," American Economic Review, 102(6), 2674–99.

- COLE, H. L., AND T. J. KEHOE (1996): "A self-fulfilling model of Mexico's 1994-1995 debt crisis," *Journal of International Economics*, 41(3-4), 309–330.
- (2000): "Self-Fulfilling Debt Crises," *Review of Economic Studies*, 67(1), 91–116.
- CONESA, J. C., AND T. J. KEHOE (2015): "Gambling for Redemption and Self-Fulfilling Debt Crises," NBER Working Papers 21026, National Bureau of Economic Research, Inc.
- CORSETTI, G., K. KUESTER, A. MEIER, AND G. J. MUELLER (2013): "Sovereign Risk, Fiscal Policy, and Macroeconomic Stability," *The Economic Journal*, 123(566), F99–F132.
- (2014): "Sovereign risk and belief-driven fluctuations in the euro area," *Journal of Monetary Economics*, 61(C), 53–73.
- CRUCES, J. J., AND C. TREBESCH (2013): "Sovereign Defaults: The Price of Haircuts," American Economic Journal: Macroeconomics, 5(3), 85–117.
- CUADRA, G., J. SANCHEZ, AND H. SAPRIZA (2010): "Fiscal Policy and Default Risk in Emerging Markets," *Review of Economic Dynamics*, 13(2), 452–469.
- DIAS, D., AND C. RICHMOND (2009): "Duration of Capital Market Exclusion: An Empirical Investigation," *Working Paper*.
- EATON, J., AND M. GERSOVITZ (1981): "Debt with Potential Repudiation: Theoretical and Empirical Analysis," *Review of Economic Studies*, 48(2), 289–309.
- FINK, F., AND A. SCHOLL (2016): "A quantitative model of sovereign debt, bailouts and conditionality," *Journal of International Economics*, 98(C), 176–190.
- FRANKEL, J. A., C. A. VEGH, AND G. VULETIN (2013): "On graduation from fiscal procyclicality," *Journal of Development Economics*, 100(1), 32–47.
- GAVIN, M., AND R. PEROTTI (1997): "Fiscal Policy in Latin America," in NBER Macroeconomics Annual 1997, Volume 12, NBER Chapters, pp. 11–72. National Bureau of Economic Research, Inc.

- GELOS, R. G., R. SAHAY, AND G. SANDLERIS (2011): "Sovereign borrowing by developing countries: What determines market access?," *Journal of International Economics*, 83(2), 243–254.
- GOMEZ-OLIVEROS DURAN, L., S. NIEMANN, AND P. PICHLER (2016): "Fiscal policy and the output costs of sovereign default," *Working Paper*.
- GONCALVES, C. E., AND B. GUIMARAES (2015): "Sovereign default risk and commitment for fiscal adjustment," *Journal of International Economics*, 95(1), 68–82.
- GOPINATH, G., AND B. NEIMAN (2014): "Trade Adjustment and Productivity in Large Crises," *American Economic Review*, 104(3), 793–831.
- GREENWOOD, J., Z. HERCOWITZ, AND G. W. HUFFMAN (1988): "Investment, Capacity Utilization, and the Real Business Cycle," *American Economic Review*, 78(3), 402–17.
- HATCHONDO, J. C., L. MARTINEZ, AND F. ROCH (2015): "Fiscal rules and the Sovereign Default Premium," Caepr Working Papers 2015-010 Classification-F, Center for Applied Economics and Policy Research, Economics Department, Indiana University Bloomington.
- HATCHONDO, J. C., L. MARTINEZ, AND H. SAPRIZA (2009): "Heterogeneous Borrowers In Quantitative Models Of Sovereign Default," *International Economic Review*, 50(4), 1129– 1151.
- ILZETZKI, E. (2011): "Rent-seeking distortions and fiscal procyclicality," Journal of Development Economics, 96(1), 30 – 46.
- JUESSEN, F., AND A. SCHABERT (2013): "Fiscal Policy, Sovereign Default, and Bailouts," IZA Discussion Papers 7805, Institute for the Study of Labor (IZA).
- KAMINSKY, G. L., C. M. REINHART, AND C. A. VEGH (2005): "When It Rains, It Pours: Procyclical Capital Flows and Macroeconomic Policies," in *NBER Macroeconomics Annual* 2004, Volume 19, NBER Chapters, pp. 11–82. National Bureau of Economic Research, Inc.
- LAHIRI, A., R. SINGH, AND C. VEGH (2007): "Segmented asset markets and optimal exchange rate regimes," *Journal of International Economics, Elsevier*, 72(1), 1–21.

- MENDOZA, E. G. (1991): "Real Business Cycles in a Small Open Economy," American Economic Review, 81(4), 797–818.
- MENDOZA, E. G., AND V. Z. YUE (2012): "A General Equilibrium Model of Sovereign Default and Business Cycles," *The Quarterly Journal of Economics*, 127(2), 889–946.
- NEUMEYER, P. A., AND F. PERRI (2005): "Business cycles in emerging economies: the role of interest rates," *Journal of Monetary Economics*, 52(2), 345–380.
- NIEMANN, S., AND P. PICHLER (2016): "Collateral, liquidity and debt sustainability," *Economic Journal, forthcoming, doi: 10.1111/ecoj.123 84.*
- POUZO, D., AND I. PRESNO (2015): "Optimal Taxation with Endogenous Default under Incomplete Markets," Papers 1508.03924, arXiv.org.
- REINHART, C. M., K. S. ROGOFF, AND M. A. SAVASTANO (2003): "Debt Intolerance," Brookings Papers on Economic Activity, 34(1), 1–74.
 - (2014): "Addicted to Dollars," Annals of Economics and Finance, 15(1), 1–50.
- SCHMITT-GROHE, S., AND M. URIBE (2007): "Optimal simple and implementable monetary and fiscal rules," *Journal of Monetary Economics*, 54(6), 1702–1725.
- SCHOLL, A. (2017): "The dynamics of sovereign default risk and political turnover," *Journal* of International Economics, 108(Supplement C), 37 – 53.
- TALVI, E., AND C. A. VEGH (2005): "Tax base variability and procyclical fiscal policy in developing countries," *Journal of Development Economics*, 78(1), 156–190.
- TAUCHEN, G. (1986): "Finite state markov-chain approximations to univariate and vector autoregressions," *Economics Letters*, 20(2), 177–181.
- TORNELL, A., AND P. R. LANE (1999): "The Voracity Effect," American Economic Review, 89(1), 22–46.
- VEGH, C. A., AND G. VULETIN (2015): "How Is Tax Policy Conducted over the Business Cycle?," *American Economic Journal: Economic Policy*, 7(3), 327–70.

A Appendix

A.1 Optimal fiscal policy

Proof of Proposition 1. We prove the converse statement that $\varepsilon \notin \Gamma(b,\xi)$ implies $\varepsilon \notin \Gamma(\tilde{b},\xi)$. Let (c, L, g) denote the equilibrium values for consumption, labor supply and public expenditure under the optimal fiscal policy $(d = 0, \tau, g, b')$ when debt is equal to b. Now consider the situation when debt is equal to $\tilde{b} < b$ and suppose the government policy remains unchanged at $(d = 0, \tau, g, b')$. From (30), bond prices are then unchanged; and since the private-sector equilibrium is completely determined by $(d = 0, \tau; \varepsilon)$, the resulting allocation then has $\tilde{c} = c$ and $\tilde{L} = L$. Hence, since $\tilde{b} < b$, the government budget constraint (21) implies that it is possible to increase public spending to $\tilde{g} > g$. Therefore, under the (possibly suboptimal) policy of keeping fiscal policy unchanged bar the residual adjustment of $\tilde{g} > g$, we have $u(\tilde{c} - v(\tilde{L}), \tilde{g}) >$ u(c - v(L), g) and $\tilde{b}' = b'$. Hence, from (22),

$$\mathcal{V}^{nd}(\tilde{b},\varepsilon) > \mathcal{V}^{nd}(b,\varepsilon).$$

Moreover, the default value $\mathcal{V}^d(\varepsilon)$ is independent of the endogenous state b. By assumption, $\varepsilon \notin \Gamma(b,\xi)$, that is,

$$\mathcal{V}^{nd}(b,\varepsilon) \ge \mathcal{V}^d(\varepsilon)$$

It then follows from the last two inequalities that $\varepsilon \notin \Gamma(\tilde{b}, \xi)$.

Proof of Proposition 2. The optimal tax policy must satisfy

$$u_c \frac{\partial c}{\partial \tau} - u_l \frac{\partial L}{\partial \tau} + u_g \frac{\partial g}{\partial \tau} = 0.$$

When the government has access to credit markets and b' > 0, its budget constraint (21) implies

$$g = \tau c + q(b', \varepsilon)b' - b,$$

whereby the bond price is determined by (30) and thus unaffected by variations in τ . Moreover,

with GHH preferences as in (1), labor supply is independent of consumption and the optimality condition for consumption-leisure (3) implies

$$\frac{u_l}{u_c} = v'(L) = \frac{w}{1+\tau}.$$

Finally, the household budget constraint (2) and the aggregation of income underlying (20) imply

$$c = \frac{1}{1+\tau} \left[wL + \pi^f + \pi^m \right] = \frac{1}{1+\tau} \left[y - P^*(r^*)m^* \right] = \frac{1}{1+\tau} gdp.$$

Using the above three equations, the optimality condition for taxes becomes

$$0 = u_c \left[\frac{\frac{\partial gdp}{\partial \tau} (1+\tau) - gdp}{(1+\tau)^2} \right] - u_c v'(L) \frac{\partial L}{\partial \tau} + u_g \left[\frac{(gdp + \tau \frac{\partial gdp}{\partial \tau})(1+\tau) - \tau gdp}{(1+\tau)^2} \right],$$

or equivalently,

$$u_c \left\{ -\frac{\partial \frac{1}{1+\tau}}{\partial \tau} g dp - \frac{1}{1+\tau} \frac{\partial g dp}{\partial \tau} + v'(L) \frac{\partial L}{\partial \tau} \right\} = u_g \left\{ -\frac{\partial \frac{1}{1+\tau}}{\partial \tau} g dp + \frac{\tau}{1+\tau} \frac{\partial g dp}{\partial \tau} \right\}.$$

From the definition of GDP in (20), $gdp = y - P^*(r^*)m^*$, and since factors earn their marginal products, while the price p_j^* for imported inputs is exogenous, we have

$$\frac{\partial g dp}{\partial \tau} = \frac{\partial g dp}{\partial L} \frac{\partial L}{\partial \tau} + \frac{\partial g dp}{\partial m^*} \frac{\partial m^*}{\partial \tau} = w \frac{\partial L}{\partial \tau} + \left[P^*(r^*) - P^*(r^*) \right] \frac{\partial m^*}{\partial \tau} = w \frac{\partial L}{\partial \tau}.$$

Hence, the optimality condition for consumption-leisure (3) implies

$$-\frac{1}{1+\tau}\frac{\partial gdp}{\partial \tau} + v'(L)\frac{\partial L}{\partial \tau} = \left[-\frac{w}{1+\tau}\frac{\partial gdp}{\partial \tau} + v'(L)\right]\frac{\partial L}{\partial \tau} = 0,$$

so that the the optimality condition for taxes becomes

$$u_c \left\{ -\frac{\partial \frac{1}{1+\tau}}{\partial \tau} g dp \right\} = u_g \left\{ -\frac{\partial \frac{1}{1+\tau}}{\partial \tau} g dp + \frac{\tau}{1+\tau} \frac{\partial g dp}{\partial \tau} \right\}.$$

It is then immediate to verify that this condition implies $u_c < u_g$.

Proof of Proposition 3. Observe first that the multiplier attached to the government budget constraint (21) is given by $u_g > 0$. Recall also (see proof of Proposition 2) that

$$c = \frac{1}{1 + \tau}gdp,$$

$$g = \tau c + q(b', \varepsilon)b' - b$$

The optimality condition for b' > 0 is then given by

$$u_g \left\{ \frac{\tau}{1+\tau} \frac{\partial g dp}{\partial b'} + q + \frac{\partial q}{\partial b'} b' \right\} + \beta E \mathcal{V}_b(b', \varepsilon', \xi') = 0,$$

where $\frac{\partial g dp}{\partial b'} = 0$ because, given the static nature of the private-sector equilibrium, the endogenous variable g dp is not affected by debt issuance. Since $\mathcal{V}^d(\varepsilon)$, the value function following default (no matter whether it occurred for fundamental or self-fulfilling reasons), does not depend on b, the envelope condition is

$$\mathcal{V}_b(b,\varepsilon) = \begin{cases} 0, & \text{if } \varepsilon \in \Gamma(b,\xi), \\ \mathcal{V}_b^{nd}(b,\varepsilon), & \text{if } \varepsilon \notin \Gamma(b,\xi), \end{cases}$$

where

$$\mathcal{V}_b^{nd}(b,\varepsilon) = u_c \left\{ \frac{1}{1+\tau} \frac{\partial g dp}{\partial b} \right\} - u_c v'(L) \frac{\partial L}{\partial b} + u_g \left\{ \frac{\tau}{1+\tau} \frac{\partial g dp}{\partial b} - 1 \right\}.$$

Since the level of maturing debt b affects the endogenous variables L and gdp not directly, but only through its impact on the government's fiscal control variables, we have $\frac{\partial L}{\partial b} = \frac{gdp}{\partial b} = 0$, and the envelope condition further simplifies to $\mathcal{V}_b^{nd}(b,\varepsilon) = -u_g$. The generalized Euler equation for b' is then obtained as

$$u_g\left\{q+\frac{\partial q}{\partial b'}b'\right\} = \beta E_{\xi'}E_{\varepsilon'\notin\Gamma(b',\xi')}u'_g.$$

From (30), bond prices are given by

$$q(b',\varepsilon) = \frac{1-\rho(b',\varepsilon)}{1+r^*} = E_{\xi'}E_{\varepsilon'\notin\Gamma(b',\xi')}\frac{1}{1+r^*}.$$

The generalized Euler equation can therefore be rewritten as

$$u_g \left\{ 1 + \epsilon_q(b') \right\} = \beta (1 + r^*) E_{\xi'} E_{\varepsilon' \notin \Gamma(b',\xi')} u'_g,$$

where $\epsilon_q(b') = \frac{\partial q}{\partial b'} \frac{b'}{q}$ denotes the elasticity of bond prices with respect to debt issuance. \Box

A.2 Business cycle statistics without rollover risk

Table 4 presents business cycle statistics for our benchmark model ($\beta = 0.94$) with and without rollover risk. As expected, the model without rollover risk generates lower spreads, higher debt

	(I)	(II)	(III)
Statistic	Data	p = 0.01	p = 0.00
Standard doviation relative	to stand	and domintic	or of CDP
	10 Stande		
C	1.20	1.28	1.16
G	1.08	1.39	1.23
τ	0.24	0.54	0.34
R_s	0.82	0.04	0.04
TB	0.65	0.47	0.29
Correlation with GDP			
C	0.93	0.95	0.98
G	0.38	0.92	0.96
au	-0.33	-0.46	-0.49
R_s	-0.64	-0.21	-0.21
TB	-0.68	-0.47	-0.48
D	-0.11	-0.13	-0.19
Correlation with spread			
C	-0.56	-0.22	-0.23
G	-0.16	-0.23	-0.23
TB	0.65	0.18	0.21
Mean debt-to-annual GDP	29%	7.6%	10.3%
Mean annual spread	4.27%	4.27%	0.14%

Table 4: Business cycle statistics with and without rollover risk ($\beta = 0.94$)

Note: The statistics reported are computed as averages over N = 2000 simulations of length T = 500, with the first 100 observations truncated. The simulated series are logged and filtered. The empirical measure for taxes is based on data for 1980-2007 and taken from Cuadra, Sanchez, and Sapriza (2010).

positions and lower relative volatilities. But the various correlations, and particularly those

for the fiscal instruments, are basically unchanged. To understand this, observe that, when $\beta = 0.94$, the level of debt is relatively close to the default threshold. Fundamentals thus place the economy within the crisis zone for most of the time. Within the crisis zone, however, rollover risk is independent of the level of debt. Accordingly, bond prices are locally sensitive to variations in fundamental default risk only, and – despite being relatively small – this shapes the cyclical properties of the optimal policies for debt issuance, taxation and spending.

Index of Working Papers:

March 5, 2015	Jonas Dovern, Martin Feldkircher, Florian Huber	200	Does Joint Modelling of the World Economy Pay Off? Evaluating Global Forecasts from a Bayesian GVAR
May 19, 2015	Markus Knell	201	The Return on Social Security with Increasing Longevity
June 15, 2015	Anil Ari	202	Sovereign Risk and Bank Risk-Taking
June 15, 2015	Matteo Crosignani	203	Why Are Banks Not Recapitalized During Crises?
February 19, 2016	Burkhard Raunig	204	Background Indicators
February 22, 2016	Jesús Crespo Cuaresma, Gernot Doppelhofer, Martin Feldkircher, Florian Huber	205	US Monetary Policy in a Globalized World
March 4, 2016	Helmut Elsinger, Philipp Schmidt- Dengler, Christine Zulehner	206	Competition in Treasury Auctions
May 14, 2016	Apostolos Thomadakis	207	Determinants of Credit Constrained Firms: Evidence from Central and Eastern Europe Region
July 1, 2016	Martin Feldkircher, Florian Huber	208	Unconventional US Monetary Policy: New Tools Same Channels?
November 24, 2016	François de Soyres	209	Value Added and Productivity Linkages Across Countries
November 25, 2016	Maria Coelho	210	Fiscal Stimulus in a Monetary Union: Evidence from Eurozone Regions
January 9, 2017	Markus Knell, Helmut Stix	211	Inequality, Perception Biases and Trust

January 31, 2017	Steve Ambler, Fabio Rumler	212	The Effectiveness of Unconventional Monetary Policy Announcements in the Euro Area: An Event and Econometric Study
May 29, 2017	Filippo De Marco	213	Bank Lending and the European Sovereign Debt Crisis
June 1, 2017	Jean-Marie Meier	214	Regulatory Integration of International Capital Markets
October 13, 2017	Markus Knell	215	Actuarial Deductions for Early Retirement
October 16, 2017	Markus Knell, Helmut Stix	216	Perceptions of Inequality
November 17, 2017	Engelbert J. Dockner, Manuel Mayer, Josef Zechner	217	Sovereign Bond Risk Premiums
December 1, 2017	Stefan Niemann, Paul Pichler	218	Optimal fiscal policy and sovereign debt crises